# A health promotion intervention to improve lifestyle choices and health outcomes in people with psychosis: a research programme including the IMPaCT RCT

Fiona Gaughran,<sup>1,2\*</sup> Daniel Stahl,<sup>3</sup> Anita Patel,<sup>4,5</sup> Khalida Ismail,<sup>6</sup> Shubulade Smith,<sup>7,8</sup> Kathryn Greenwood,<sup>9,10</sup> Zerrin Atakan,<sup>2</sup> Poonam Gardner-Sood,<sup>2</sup> Dominic Stringer,<sup>3</sup> David Hopkins,<sup>11</sup> John Lally,<sup>1,2,12</sup> Marta Di Forti,<sup>13,14,15</sup> Brendon Stubbs,<sup>16,17</sup> Philippa Lowe,<sup>18</sup> Maurice Arbuthnott,<sup>19</sup> Margaret Heslin,<sup>20</sup> Anthony S David<sup>21</sup> and Robin M Murray<sup>2,14,15</sup>

- <sup>1</sup>National Psychosis Service, South London and Maudsley NHS Foundation Trust, London, UK
- <sup>2</sup>Department of Psychosis Studies, Institute of Psychiatry, Psychology & Neuroscience, King's College London, London, UK
- <sup>3</sup>Department of Biostatistics and Health Informatics, Institute of Psychiatry, Psychology & Neuroscience, King's College London, London, UK
- <sup>4</sup>Anita Patel Health Economics Consulting Ltd, London, UK
- <sup>5</sup>Centre for Primary Care and Public Health, Blizard Institute, Queen Mary University of London, London, UK
- <sup>6</sup>Department of Psychological Medicine, Institute of Psychiatry, Psychology & Neuroscience, King's College London, London, UK
- <sup>7</sup>Department of Forensic and Neurodevelopmental Science, Institute of Psychiatry, Psychology & Neuroscience, King's College London, London, UK
- <sup>8</sup>Forensic Services, South London and Maudsley NHS Foundation Trust, London, UK <sup>9</sup>Sussex Partnership NHS Foundation Trust, Worthing, UK
- <sup>10</sup>School of Psychology, University of Sussex, Brighton, UK
- <sup>11</sup>Institute of Diabetes, Endocrinology and Obesity, King's Health Partners, London, UK
- <sup>12</sup>Department of Psychiatry, Royal College of Surgeons in Ireland, Beaumont Hospital, Dublin, Ireland
- <sup>13</sup>Social, Genetic & Developmental Psychiatry Centre, Institute of Psychiatry, Psychology & Neuroscience, King's College, London, UK
- <sup>14</sup>Department of Psychiatry, Experimental Biomedicine and Clinical Neuroscience (BIONEC), University of Palermo, Palermo, Italy
- <sup>15</sup>South London and Maudsley NHS Foundation Trust, London, UK

- <sup>16</sup>Department of Psychological Medicine, Institute of Psychiatry, Psychology & Neuroscience, King's College London, London, UK
- <sup>17</sup>Physiotherapy Department, South London and Maudsley NHS Foundation Trust, London, UK
- <sup>18</sup>Carer expert and Rethink Mental Illness, London, UK
- <sup>19</sup>Service user expert, London, UK
- <sup>20</sup>King's Health Economics, Health Service & Population Research Department, Institute of Psychiatry, Psychology & Neuroscience, King's College London, London, UK
- <sup>21</sup>Institute of Mental Health, Division of Psychiatry, University College London, London, UK

\*Corresponding author fiona.p.gaughran@kcl.ac.uk

**Declared competing interests of authors:** Fiona Gaughran reports support from Janssen Pharmaceutica (Johnson & Johnson, Beerse, Belgium), personal fees from Sunovion Pharmaceuticals Inc. (Marlborough, MA, USA), H. Lundbeck A/S (Copenhagen, Denmark), F. Hoffman-La Roche Ltd (Basel, Switzerland) and Otsuka Pharmaceutical Co., Ltd (Tokyo, Japan) outside the submitted work; and has a family member with professional links to Eli Lilly and Company (Indianapolis, IN, USA) and to GlaxoSmithKline plc (London, UK), including shares. She is in part supported by the Maudsley Charity and the National Institute for Health Research (NIHR) Applied Research Collaboration South London at King's College Hospital NHS Foundation Trust. Khalida Ismail has been paid honorarium by Eli Lilly and Company, Janssen Pharmaceutica, Sanofi S.A. (Paris, France) and Novo Nordisk A/S (Bagsværd, Denmark) for educational lectures. David Hopkins reports personal fees from Sunovion Pharmaceuticals Inc., Eli Lilly and Company, Novo Nordisk A/S, AstraZeneca plc (Cambridge, UK), F. Hoffman-La Roche Ltd, Medtronic plc (Dublin, Ireland), Fractyl, Inc (Lexington MA, USA) and Sanofi, outside the submitted work. Robin M Murray has received honoraria for lectures from Lundbeck, Otsuka, Janssen and Sunovian.

Published January 2020 DOI: 10.3310/pgfar08010

# **Plain English summary**

## The IMPaCT RCT

Programme Grants for Applied Research 2020; Vol. 8: No. 1 DOI: 10.3310/pgfar08010

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

### Background

People with psychosis have poor physical health and reduced life expectancy, with cardiovascular disease causing about 60% of early deaths in this population.

#### Aims

We aimed to record the physical health risks and lifestyle choices of those with early psychosis, devise a way to help people make healthier choices and test whether or not this improved their quality of life.

#### **Methods**

First, we followed people for the year after they first presented with psychosis, documenting their mental and physical health and their lifestyle choices. Second, we then worked with carers and service users to develop a health promotion intervention, IMPaCT therapy, to address exercise, diet, smoking, alcohol and substance use, plus diabetes management where relevant. Finally, we ran a trial to see whether or not IMPaCT therapy helped and if it was good value. The work took place between 2008 and 2014.

#### Results

Patients with their first episode of psychosis had many risk factors for heart disease, such as obesity, diabetes, high levels of cholesterol, poor diet and little exercise.

In the IMPaCT therapy trial, we recruited over 100 care co-ordinators and split them into two groups randomly; half were trained in IMPaCT therapy and offered supervision of their practice, and the other half provided standard care. We reassessed the patients 1 year later to see whether or not there was any difference in quality of life and physical or mental health. We found no extra benefit for the patients allocated to IMPaCT therapy compared with those receiving standard care, although if at least 3 hours of IMPaCT therapy was received on top of standard care, waist circumference reduced. Costs of the intervention were low.

These findings signify the difficulty in reducing risk factors for physical health problems once these are established.

## **Programme Grants for Applied Research**

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **Programme Grants for Applied Research programme**

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

#### This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-0606-1049. The contractual start date was in August 2007. The final report began editorial review in September 2018 and was accepted for publication in September 2019. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGfAR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Gaughran *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

© Queen's Printer and Controller of HMSO 2020. This work was produced by Gaughran *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

### Editor-in-Chief of Programme Grants for Applied Research and NIHR Journals Library

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

### NIHR Journals Library Editors

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk